Author's response to reviews

Title: Clinical presentation of human African trypanosomiasis in Zambia links with varying virulent strains of human trypanosomes: Two case reports.

Authors:

Victor Mwanakasale (vicsale@zamtel.zm)
Peter Songolo (songolop@zm.afro.who.int)
Olusegun Babaniyi (babaniyio@zm.afro.who.int)
Pere Simarro (SimarroP@who.int)

Version: 3
Date: 12 November 2013

Author's response to reviews: see over
Point by point responses to Editorial Reviewers’ comments.

1. Please change the description of the patient’s gender from woman to female.  
   Response: This has been changed.

2. Please change the description of the patient’s gender from man to male.  
   Response: This has been changed.

3. Please include the ethnicity of the patient in the abstract and case presentation sections.  
   Response: This has been done.

4. Please change the ‘background’ section header to ‘Introduction’.  
   Response: This has been done.

5. Please remove the dates in the Case presentation section.  
   Response: This has been done.

6. Remove the hospital name in the Case presentation as this may jeopardize the anonymity of the patient.  
   Response: This has been done.

7. After reading through the manuscript, we feel that the quality of written English needs to be improved.  
   Response: This has been addressed. We hope the quality is fine now.
Point by point respondes to Specific reviewers comments:

Maria de nazare Correia MN Soeiro.

1. English.
   Response: This has been addressed.

2. Abstract.
   Response: This has been revised

3. Key words.
   Response: This has been done.

4. Follow dates.
   Response: Dates have been deleted from the script

James CL Mwansa.

1. No response

2. Assumed virulence.
   Response: Clarified

3. Laboratory findings after death or treatment.
   Response: Not done

4. Implication to national control.
   Response: Clarified

5. Duration of treatment rather than dates.
   Response: Dates are appropriate in treatment of Human African Trypanosomiasis.

   Response: This has been changed.